Author's response to reviews

Title: Predictors of neurologic outcome after successful extracorporeal cardiopulmonary resuscitation

Authors:

Jeong-Am Ryu (lamyud.ryu@samsung.com)
Yang Hyun Cho (yanghyun.cho@samsung.com)
Kiick Sung (kiick.sung@samsung.com)
Seung Hyuk Choi (sh1214.choi@samsung.com)
Jeong Hoon Yang (jeonghoon.yang@samsung.com)
Jin-Ho Choi (jin-ho.choi@samsung.com)
Ji-Hyuk Yang (jh1.yang@samsung.com)

Version: 5 Date: 5 November 2014

Author's response to reviews:

Dear Dr. Tom Rowles

We have recently obtained some data on extracorporeal cardiopulmonary resuscitation (ECPR). The title of our manuscript is “Predictors of neurologic outcome after successful ECPR”. We are sending here with an original manuscript.

The paper has been written according to the guidelines provide by BMC Anesthesiology. I would appreciate very much if you could review it for publication in BMC Anesthesiology. Neither the entire paper nor any part of its content has been published of has been accepted by another journal. We believe the paper may be of particular interest to your reader because the study reports a new and interesting subjective of neurologic outcome after ECPR.

Correspondence and phone calls about the paper should be directed to me at the following address and phone number:

Ji-Hyuk Yang MD, PhD
Department of Thoracic and Cardiovascular Surgery, Samsung Medical Center, Sungkyunkwan University School of Medicine, 81 Irwon-ro, Gangam-gu, Seoul 135-710, Republic of Korea.
Tel: 82-2-3410-5065, Fax: 82-2-3410-6986
E-mail: jh1.yang@samsung.com

We thank you in advance for your concern.

Sincerely your,
Ji-Hyuk Yang MD, PhD
Professor